

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Dohmen, Thomas J.

Working Paper
Housing, Mobility and Unemployment

IZA Discussion Papers, No. 210

Provided in Cooperation with:

IZA - Institute of Labor Economics

Suggested Citation: Dohmen, Thomas J. (2000): Housing, Mobility and Unemployment, IZA Discussion Papers, No. 210, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/21055

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





IZA DP No. 210

Housing, Mobility and Unemployment

Thomas J. Dohmen

November 2000

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Housing, Mobility and Unemployment

Thomas J. Dohmen

University of Maastricht, Business Investment Research Center and IZA, Bonn

> Discussion Paper No. 210 November 2000

> > IZA

P.O. Box 7240 D-53072 Bonn Germany

Tel.: +49-228-3894-0 Fax: +49-228-3894-210 Email: iza@iza.org

This Discussion Paper is issued within the framework of IZA's research area *Mobility and Flexibility of Labor Markets*. Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent, nonprofit limited liability company (Gesellschaft mit beschränkter Haftung) supported by the Deutsche Post AG. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public. The current research program deals with (1) mobility and flexibility of labor markets, (2) internationalization of labor markets and European integration, (3) the welfare state and labor markets, (4) labor markets in transition, (5) the future of work, (6) project evaluation and (7) general labor economics.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character.

ABSTRACT

Housing, Mobility and Unemployment*

This paper develops a model that shows why high-skilled workers move more and are therefore unemployed less than low-skilled workers. The model can explain the paradoxical empirical regularity that higher owner-occupation rates are associated with higher levels of unemployment although home-owners tend to be unemployed less. The choice of housing tenure affects moving costs and thereby regional mobility and unemployment. The paper analyzes the impact of symmetric and asymmetric shocks on mobility and unemployment, and discusses effects of government intervention in the housing market. In addition, it is shown that moving costs reduce job search effort and search effectiveness.

JEL Classification: J6

Keywords: Housing, mobility, unemployment

Thomas J. Dohmen Business Investment Research Center University of Maastricht P.O. Box 616 6200 MD MAASTRICHT The Netherlands

Tel.: +31-43-388 3832 Fax: +31-43-388 4856

Email: t.dohmen@mw.unimaas.nl

* I thank Andrew Oswald, Gerard Pfann and Ben Kriechel for fruitful discussions and very useful comments.

1 Introduction

Despite the remarkable progress that has been made over in the past decade in developing theories to explain unemployment, the puzzle of high European unemployment remains largely unsolved. Blanchard and Katz (1997) assert: "The magnitude of the increase [in unemployment], the diversity of labor market institutions and of policies should have helped identify culprits. Yet while many suspects have been identified, none has been convicted." Consistent with this statement is the result of a recent paper by Nickell (1997) which does not find a satisfactory empirical explanation when estimating how labor market rigidities, the treatment of he unemployed, taxes, union coverage, union and employer co-ordination in wage bargaining – all key factors in prominent models – impact on unemployment. Nickell (1997) remarks that both geographical and occupational mobility of workers seem to be higher in the United States than in Europe where unemployment is higher. But he does not explore its causes any further, nor does he discuss the dimension of the role mobility could play as an explanatory variable in unemployment equations.

At the same time, empirical work suggests that housing plays an important role in the determination of unemployment.¹ When Nickell (1998) adds the proportion of home-ownership as an explanatory variable of unemployment rates, the fit of his regression improves considerably. However, compared to empirical work, little effort has been devoted to develop models that explain regional mobility and to enlarge existing models of unemployment, although the idea that mobility adversely affects labor market outcomes is at least as old as the idea of the natural rate of unemployment.²

This paper tries to fill this gap. It develops a model which studies the link between skills, the choice of housing tenure and mobility. It explains why high-skilled workers are more mobile than low-skilled workers given fixed moving costs. Moreover it shows how fixed moving costs impede geographical mobility and therefore affect labor market outcomes by reducing search intensity, diminishing the attractiveness of job offers, hence lowering wage pressure and increasing long-term unemployment. The model generates pre-

¹See Hughes and McCormick (1981), (1985a), (1987), Oswald (1996), (1997b), Owen and Green (1997) and Henley (1998).

²Friedman himself explained: "The natural rate of unemployment is the level which would be ground out by the Walrasian system of general equilibrium equations, provided that there is imbedded in them the actual structural characteristics of the labor and commodity markets, including market imperfections, stochastic variability in demands and supplies, the cost of gathering information about job vacancies and labor availabilities, the cost of mobility, and so on."

dictions consistent with empirical results and stylized facts, and yields important implications for policy. The paper does not attempt to develop a new theory of unemployment but rather to arrange some simple ideas in a logical way that can readily be integrated into existing models.

The paper is organized as follows. The following section discusses links between housing and labor markets and reports some stylized facts that motivate the construction of a simple model of housing, mobility and unemployment in section 3. Section 4 discusses how shocks affect mobility and unemployment, while section 5 introduces some refinements of the simple model. Sections 6 and 7 extend the model to analyze the interaction between search and moving costs as well as the consequences of stigmatization and insider forces. Section 8 focuses on policy implication. Section 9 proposes possible extensions of the model for future work. Finally, section 10 concludes.

2 Links between Housing and Labor Markets - Stylized Facts and Empirical Evidence

There is a growing body of evidence which suggest a positive correlation between rates of home-ownership and unemployment. Oswald (1996), for example, finds that the gradient of unemployment to owner-occupation is roughly 0.2 in different data sets including cross-sections of OECD countries, UK regions, U.S. states, regions of France, Sweden and Italy and a panel data set of UK regions. This empirical regularity is buttressed by the fact that nations with fastest growth rates had fastest growth in unemployment. Again, a 10 percentage point increase in the rate of owner-occupation is associated with an increase in the unemployment rate of approximately 2 percentage points. Higher rates of owner-occupation seem to impede mobility and hence raise unemployment. The rate of owner-occupation might therefore be an important factor omitted in empirical studies of unemployment. In fact, the fit of Nickell's (1997) regression to explain equilibrium unemployment improves considerably – the R²rises from .76 to .82 – when he adds the proportion of home- ownership as a proxy for mobility to the list of explanatory variables. The estimated coefficient of this variable implies that a 10 percentage point rise in the owner-occupation rate is associated with an additional 1.3 percentage points of the unemployment rate.³

³It should be noted that the equation is overfitted – there are only 28 degrees of freedom as 12 coefficients are estimated on 40 observations – such that not too much weight should be given to the point estimate of the coefficient on owner-occupation. It is clearly possible

The idea that housing and labor markets are linked is not new, however. Research by Hughes and McCormick (1981, 1985a, 1987) focuses on the private and public renting sector and concludes that council houses restrict labor mobility and hence raise unemployment. Bover et al. (1989) argue that regional house price/earnings differentials are an important element in wage and unemployment, vacancy equations. More recent papers have studied differences in mobility of owners and renters. According to Owen and Green (1997), private renters move relatively more than owners. Oswald (1997b) cites statistics from the 1995 General Household survey which indicates that renters have lived for shorter periods at the current address. Only 5% of all owners have lived at their current address for less than a year, while 27% of renters in the private unfurnished sector and 49% in the furnished sector moved in during the last 12 months. Henley (1998) reports findings suggesting that home ownership reduces mobility.

In spite of the empirical evidence there seems to be a lack of effort to develop theoretical models that are capable of explaining the process generating the evidence in the data. An exception is Oswald (1997a) who develops a model which predicts that higher rates of home-ownership are associated with more unemployment and that unemployment is concentrated among home-owners. However, unemployment data does not suggest that renters are unemployed less. Problems arise also when the model is extended in sensible ways; for example, when one assumes that skilled workers who have higher incomes than unskilled are more likely to own a house. According to the model unskilled are expected to move more and to be unemployed less frequently since they are more likely to be renters and therefore more mobile. Again, this is not what the data suggests.

The model developed in this paper explains how higher rates of home ownership can lead to higher levels of unemployment while unemployment is concentrated among low-skilled workers and possibly among renters as high-skilled workers are more mobile than low-skilled workers given fixed moving costs. The model generates predictions consistent with a number of empirical results and stylized facts concerning labor market outcomes which also motivate the building of the model.

First, flows in the labor markets are immense. Workers quit their job to search for better ones or to leave the labor force, others are fired, releasing a new wave of replacements. Yet another group enters the labor market. The magnitude of those gross flows is much higher than net flows might suggest and strikingly similar in magnitude across OECD countries (for evidence

that the gradient of unemployment to owner-occupation is larger and closer to 0.2 in a more parsimonious specification.

see Davis et. al, 1996; Contini and Revelli, 1997; OECD, 1996, chapter 5). Second, most of the rise in European unemployment can be attributed to an increase in duration hence a rise of the proportion of long-term unemployment (see Nickell, 1997) rather than to a higher inflow rate. Furthermore, most of the differences in unemployment rates across countries can be ascribed to variations in long-term unemployment. Short-term unemployment seems to be quite similar across countries, indicating that the amount of frictional unemployment – which does exist in equilibrium due to the reallocation of workers – is quite similar. Together with the third empirical regularity, that rates of home-ownership are positively correlated with unemployment, this suggests that mobility or better impediments to mobility play a crucial role in the determination of unemployment. It becomes harder to match workers and firms if workers are less prepared to move regions and spend less effort on search across regions. The consequent fall in search effectiveness is consistent with the fourth stylized fact: the outward shift of the Beveridge curve. In most countries, the number of unemployed has risen relative to the number of vacancies indicating that vacancies are filled less efficiently.

Fifth, there are important differences in unemployment rates between demographic and occupational groups. Skilled workers are unemployed less and have shorter expected unemployment durations. Sixth, skilled workers move more, search more and have a preference for employed search. Hughes and McCormick (1985b) find that education increases the probability of job search and Pissarides and Wadsworth (1994) provide evidence that skilled workers search more and prefer employed search. McCormick (1997) argues that unemployment rates are largely determined in the manual labor market and finds little evidence that manual labor moves to low unemployment markets. Instead, wide variations in regional unemployment rates exist among manual workers, whereas absolute net migration rates are much higher for non-manuals. Evans and McCormick (1994) distinguish in their empirical study between manual and non-manual workers and cite evidence that the market for non-manual labor is flexible and geographically integrated while the manual labor market is spatially rigid.

The model developed in this paper can be integrated into existing models in a rather straightforward fashion. Reduced mobility and search activity lead to a fall in search effectiveness and hence a reduction of wage pressure. If mobility and search behaviour are partly determined by conditions in the housing market, the wage pressure variable in models of the Layard and Nickell type (see Layard et al., 1991) or the search effectiveness parameter in matching models (see, for example, Pissarides, 1990) becomes a function of these conditions. Hence, interventions in the housing markets would affect labor market outcomes. This has important policy implications.

3 A Simple Model

This section introduces a simple model which is consistent with the empirical regularity that higher rates of home-ownership are associated with higher rates of unemployment. Moreover, it explains why high-skilled workers move more and are less often unemployed. A condition is derived under which high-skilled home-owners are more likely to move than low-skilled renters. The model is based on the following assumptions:

- 1. There are two identical regions, East and West. Hence, the size of the regions and the number of jobs in both regions are the same.
- 2. Workers must live where their job is (unemployed can live anywhere). Commuting is not feasible. If a worker accepts a job offer in the region where he does not live, he must move and incur the fixed moving costs k^m .
- 3. Workers are either home-owners or renters. The fixed moving costs are equal to k^o for owners, while renters incur a fixed cost of k^r when they change the region of residency. Moving is more expensive for owners, hence $k^o > k^r$.
- 4. There are two types of workers: High-skilled workers who earn a wage H and low-skilled workers who earn L, where H > L. Hence, a worker's wage, W, is defined as W = H if the worker is high-skilled and W = L if the worker is low-skilled.
- 5. For simplicity, it is assumed that high-skilled workers only receive job offers for skilled jobs and low-skilled workers only offers for unskilled jobs.⁴
- 6. Jobs last for one period. At the end of the period each worker receives one job offer. The offer comes from a firm in West with probability q and from a firm in East with probability (1-q). Workers can either accept and earn W or reject. If they reject they become unemployed for one period and get another job offer at the end of the next period.
- 7. In absence of asymmetric shocks, q is assumed to equal 0.5 such that a job offer comes from either region with equal probability.
- 8. Unemployed receive benefits B with B < L. Benefits are net of any costs of unemployment, C, like mental costs. Gross benefits include any payment or subsidy an unemployed worker receives as well as the monetary value of additional leisure time or self-employment he enjoys. Let M be the monetary value of gross benefits and C the monetary value of any cost associated with unemployment, then B = M C.
- 9. The discount factor is equal to δ , $0 < \delta < 1$.
- 10. Worker's lives are infinite.

⁴Equivalently, we could assume that high-skilled workers receive a job offer for a skilled and an unskilled job, but that the condition that would make high-skilled workers prefer a low-skilled job to a high-skilled job is never satisfied, i.e. that $H > L + k^m$.

The last assumption is not crucial, since a finite horizon is generally sufficient to derive the important implications of the model. However, an infinite horizon sometimes facilitates the exposition. Assumption 6 is motivated by high rates of destruction.⁵ Job destruction rates determine the expected length of an employment spell. To approximate this, it is assumed that employment lasts for a given period of time.⁶ The assumption that employment lasts as long as the time between two job offers makes the mathematics much more transparent without affecting the main insights of the model. The predictions of the model continue to hold when employment is assumed to last longer than the interval between two job offers.

<u>Proposition 1:</u> Workers always accept job offers in their home region, i.e. <u>employment is preferred to unemployment ceteris paribus.</u>

Proof: H > B and L > B by assumptions 4 and 8.

<u>Proposition 2:</u> Given moving costs, high-skilled workers are more likely to move than low-skilled workers. There is a condition under which high-skilled workers always move if the offer comes from the other region while low-skilled workers never move. In that case, high-skilled workers never become unemployed, while low-skilled workers who do not receive a job offer where they live always become unemployed.

Proof: Suppose a worker receives a job offer in the other region. If he accepts and moves, his expected income, Y^a , is given by:

$$Y^{a} = W - k^{m} + \frac{0.5\delta}{1 - \delta}(W + \max[W - k^{m}; B])$$

his expected income when rejecting, Y^r , and staying is given by:

$$Y^{r} = B + \frac{0.5\delta}{1 - \delta}(W + \max[W - k^{m}; B]).$$

(Note that q equals 0.5 by assumption 7.) Hence, a worker moves if

$$W - B > k^m. (1)$$

⁵See Davis et al. (1996), for example.

⁶This simplification is made despite being aware that there is great variation in employment spells. Farber (1999) summarizes some of the evidence. The assumption is not problematic, however, since it is shown in later sections, how mobility and unemployment are affected when expected lengths of employment spell differ across individuals.

Since H - B > L - B by assumption 4, a high-skilled worker is more likely to satisfy (1) than a low-skilled worker, given moving costs. A high-skilled worker moves if

$$H - B > k^m, (2)$$

a low-skilled worker prefers moving to unemployment if

$$L - B > k^m. (3)$$

All workers move if (3) holds, whereas nobody moves if

$$H < B + k^m. (4)$$

Only high-skilled workers move if (2) holds but (3) does not, i.e. if

$$H > B + k^m > L. (5)$$

High-skilled workers are more mobile in this case and will not become unemployed while less-skilled do.

<u>Proposition 3:</u> A rise in net benefits, B, or an increase in moving costs make a worker of any type less likely to move and therefore more likely to become unemployed.

Proof: A rise in B reduces the excess value of employment and makes a worker less likely to satisfy (1). If moving costs, k^m , increase, a larger excess value of employment relative to unemployment, W - B, is required to make a worker move. Again, a worker is less likely to satisfy (1).

<u>Proposition 4:</u> Renters are more likely to move than home-owners, ceteris paribus. Therefore, a rise in the rate of home-ownership results in rising average moving costs and a fall in the acceptance rate of job offers. Hence, unemployment rises.

Proof: By assumption 3, moving costs are higher for home-owners, i.e. $k^o > k^r$. Hence, ceteris paribus, home-owners are less likely to move than

⁷The result is the same if unemployment benefits are a proportion of the last wage as is common in numerous countries. M is then equal to bW+S, where b is the replacement rate and S are other benefits associated with unemployment. The moving condition can hence be written as $W > (S-C+k^m)/(1-b)$, and a rise in b or an increase in moving costs increases mobility and hence reduces the expected duration of unemployment. The condition under which high-skilled always move while low-skilled rather go through a spell of unemployment is given by $H > (S-C+k^m)/(1-b) > L$ (1.4a)

renters.

<u>Proposition 5:</u> High-skilled renters are most likely to move, whereas low-skilled owners are least likely to move. High-skilled owners are more mobile than less-skilled renters if the difference between high-skilled and low-skilled wages exceeds the difference between moving costs for renters and owners, i.e. if $H - L > k^o - k^r$.

Proof: It follows from propositions 2 and 4 that high-skilled renters are most likely to move and low-skilled owners least likely to move. High-skilled owners move if

$$H - B > k^o, (6)$$

low-skilled renters move if

$$L - B > k^r. (7)$$

High-skilled owners are more mobile than low-skilled renters if their net benefit from moving exceeds the net benefit that low-skilled renters enjoy, i.e. if

$$H - B - k^{o} > L - B - k^{r}$$

$$\Leftrightarrow H - L > k^{o} - k^{r}.$$
(8)

This simple model predicts that regional mobility depends on wages relative to moving costs and unemployment benefits. A rise in benefits or moving costs reduces mobility. Workers with higher expected wages are more mobile and therefore less likely to become unemployed, for given moving costs and unemployment benefits. Higher education usually results in higher wages and therefore higher mobility. In addition, higher educated workers may incur lower fixed moving costs. Home-owners incur higher moving costs and are therefore less likely to move. However, if workers with higher wages are more likely to be home-owners, we might observe that home-owners are on average more mobile if we do not correct for income. An increase in the proportion of less-skilled (high-skilled) owners unambiguously has an adverse effect on employment if moving is not optimal for less-skilled (high-skilled) owners but for less-skilled (high-skilled) renters.

Since the proportion of ownership is, in equilibrium, determined by the condition that the utility from owning is equal to the utility of renting, a

⁸It is likely that higher educated workers have lower psychic moving costs. First, those with university degree are somewhat used to move regions. Second, endogenous selection might be present as those who have strong aversions towards moving are likely not to opt for higher education. Third, information about job vacancies in other regions might be easier to obtain for high-skilled. Fourth, firms that offer jobs to higher educated workers sometimes pay physical moving costs.

change in the proportion can only be brought about by changes in relatives costs and benefits of owning which can be induced by changes in tastes (preferences) or government intervention. If owning is a luxury good, such that an increase in the proportion of home-owners is brought about by an exogenous increase in wages, for example due to a productivity increase, then the effect on average mobility is ambiguous. Although average moving costs rise, incomes relative to moving costs may rise or fall.

It seems rather unrealistic that, for example, low-skilled renters either always or never move, or that lower moving costs have no impact on mobility over some range but cause discrete jumps in mobility rates when making a moving constraint non-binding. These discrete outcomes of the model obviously follow from the assumptions. This should not bother the reader. The discreteness disappear if we either assume a continuum of possible wage offers for different workers (in reality there is a continuum of skills and worker productivities) or a continuum of mobility costs. A continuous distribution of individual moving costs is realistic as a number of factors contribute to moving costs, for example, marital status, number of children, age of children, individual costs of moving (importance of friends, neighborhood, etc.). Changing the model to allow for a continuum of moving costs is a straightforward exercise. The model explains the composition of unemployment. The level of unemployment is only partly determined endogenously, namely to the extent that unemployment is caused by lack of mobility. However, other factors that are exogenous to the model might influence the level of unemployment. For example, a lower job offer arrival rate results in a higher level of unemployment as the expected duration of an unemployment spell increases. The next section will, inter alia, extend the model in that direction. The focus of the next section is also on the impact of shocks which affect the two regions differently.

4 The Effects of Shocks

This section analyzes the impact of shocks on mobility. It is shown how symmetric and asymmetric shocks impact on unemployment and how they affect different types of workers. An adverse symmetric shock is modeled as a fall of the job offer arrival rate. In particular, it is assumed that workers receive a job offer every second period rather than every period. Analogously, a positive symmetric shock is modeled as an increase in the job offer arrival

⁹The utility derived from owning might differ across countries mostly because of cultural reasons. For an extensive treatment of changes in preferences see for example Oswald (1996).

rate.

4.1 Asymmetric Shocks

Proposition 6: A temporary random asymmetric shock which reduces employment opportunities in one region (say East) and increases job offers and employment opportunities in the other region (West) only affects workers who satisfy the moving constraint (1) since the shock does not alter moving costs. It tends to increase the number of workers moving to West and lower the number of workers moving to East. An asymmetric shock which raises employment in West and lowers employment in East and is expected to persist in the future makes workers in East more likely to move and workers in West, the booming region, more likely to stay. The effect is bigger, the longer the shock is expected to persist. Therefore, serially correlated shocks have a bigger impact on mobility than random shocks.

Proof:

I) Random, serially uncorrelated temporary shocks

An asymmetric shock which increases job offers in West raises the probability q of a job offer in West. If the shock is temporary, say lasts one period, q exceeds one half only in the current period. The moving constraint is still given by (1) since the expected income when moving, for those who get a job offer in the other region, is given by:

$$W - k^m + \frac{0.5\delta}{1 - \delta}(W + \max[W - k^m; B])$$

whereas the expected income when rejecting and staying is given by:

$$B + \frac{0.5\delta}{1 - \delta}(W + \max[W - k^m; B]).$$

In absence of the shock, half of the workers who do not satisfy (1) receive a job offer in the other region and become unemployed because they do not move. Because of the shock, more workers in East are offered a job in West while less workers in West are offered a job in East. Those who do not satisfy (1) become unemployed. Thus, unemployment rises in East and falls in West. Unemployment is concentrated among the immobile workers.

II) <u>Persistent Shocks</u>

a) First consider a worker in East, the slump region, who receives an offer from West. For simplicity, assume that the difference in the rate of job offers persists only in the next future period. His expected discounted future

income when moving is given by:

$$W - k^{m} + \delta(qW + (1 - q) * \max[W - k^{m}; B]) + \frac{0.5\delta^{2}}{1 - \delta}(W + \max[W - k^{m}; B]),$$

the present value of staying is given by

$$B + \delta((1 - q)W + q * max[W - k^m; B]) + \frac{0.5\delta^2}{1 - \delta}(W + max[W - k^m; B]).$$

Hence, moving is profitable if

$$W - B > k^m - \delta((2q - 1)W - (2q - 1) * \max[W - k^m; B]). \tag{9}$$

For workers satisfying (1) this can be written as

$$W - B > k^m - \delta(2q - 1)k^m \tag{10}$$

For workers not satisfying (1) the new moving constraint (9) is given by

$$W - B > k^{m} - \delta(2q - 1)(W - B)$$

$$\Leftrightarrow W - B > \frac{k^{m}}{1 + \delta(2q - 1)}$$

$$\tag{11}$$

The term in brackets on the right-hand side of (9) is always positive since by assumption $W > B, k^m > 0$ and q, the probability of receiving a job offer from in West, exceeds 0.5. Comparing (9) to (1) and (4), it is obvious that workers in East are more likely to accept a job offer in West and move, as long as the job offer arrival rate remains higher in West at least throughout the next period.

b) Next, consider a worker in West, the booming region, who receives an offer from East. For simplicity, assume that the difference in job offers persists only in the next future period: His expected discounted future income when moving is given by

$$W - k^{m} + \delta((1 - q)W + q * \max[W - k^{m}; B]) + \frac{0.5\delta^{2}}{1 - \delta}(W + \max[W - k^{m}; B]),$$

the present value of staying is given by

$$\begin{split} B + \delta(qW + (1-q)*max[W - k^m; B]) \\ + \frac{0.5\delta^2}{1-\delta}(W + max[W - k^m; B]), \end{split}$$

Hence, moving is profitable if

$$W - B > k^m + \delta \{(2q - 1)W - (2q - 1) * \max[W - k^m; B]\}$$
 (12)

This can be written, for workers satisfying (1), as

$$W - B > k^m + \delta(2q - 1)k^m \tag{13}$$

and for workers not satisfying (1) as

$$W - B > k^m + \delta(2q - 1)(W - B). \tag{14}$$

Since 2q - 1 > 0, the difference between the wage and the unemployment benefits must be higher to make moving worthwhile than in conditions (1).

If the shock is serially correlated, the moving condition is given by

$$W - B > k^m - 2\delta\rho\sigma k^m,\tag{15}$$

where ρ is defined as the correlation coefficient and σ is a measure of the shock with $0 < \sigma < 0.5$ and $0 < \rho < 1$. This is derived as follows: A worker in the depressed region who accepts a job offer in the booming region and who will always move, has an expected income of

$$W - k^m + \delta(W - 0.5k^m + \rho\sigma k^m) + \delta^2(W - 0.5k^m + \rho^3\sigma^3 k^m) + \dots$$

Somebody who rejects, but would move in the second period receives

$$B + \delta(W - 0.5k^m - \rho\sigma k^m) + \delta^2(W - 0.5k^m + \rho^3\sigma^3k^m) + \dots$$
 (16)

Notice that expected income from the second period onwards is the same as the probability of being in the booming region is the same. Hence, moving is profitable if

$$W - B > k^m - 2\delta\rho\sigma k^m$$
.

The bigger (is, i.e. the more workers value the future, and the bigger q is, i.e. the more intense the shock is, and the longer its persistence (for a proof see the appendix), the more attractive moving becomes for workers in East and staying for workers in West, the booming region. Equation (9) shows that the moving constraint is eased more the higher the wage is. Those who are least likely to move in absence of a shock are therefore

¹⁰It should be noted that moving costs to the booming region may increase for owners if they find it harder to sell their home in the depressed region; on the other hand, costs are also likely to rise for renters in a perfect market as rents in the booming region are expected to increase.

the last to satisfy (9). So, the model predicts that higher skilled workers are the first to leave a depressed region. This effect is magnified if there is uncertainty about the length of the shock. To appreciate this argument, assume that shocks either persist 3 periods or 1 period. Furthermore, workers cannot observe the type of shock initially. If both shocks are equally likely to occur, the expected length of a shock is 2 periods. Suppose that there is a group for which moving is optimal only if the shock lasts at least for 3 periods. Given an expected duration of 2 periods, it is ex ante never optimal for them to move, although in half of the cases moving is optimal When workers learn about the true type of the shock (after 1) period say), the remaining duration of the shock is too short to make a move profitable. The moving constraint (9) might remain binding for some workers while unemployment in the depressed region is increasing¹¹. This would be consistent with Evans and McCormick's (1994) finding that relative demand shocks do not stimulate out-migration from the depressed region unless local wages fall. 12 We would then conclude that migration rates for a given group of workers do not respond to higher unemployment rates. Yet, higher unemployment has an (indirect) effect on regional mobility if wages are flexible. If higher unemployment creates downward pressure on wages, relative wages will rise in the booming region, West. Moving away from the depressed region, East, becomes more attractive. To illustrate, consider again a worker in East, the depressed region, who gets a job offer in the booming region, West, which pays W^W . Assume that a job in East pays W^E , where $W^W > W > W^E$, and that the shock lasts for one future period, after which wages are equal again in both regions. His expected discounted future income when moving to West is given by

$$\begin{split} W^W - k^m + \delta \{qW^W + (1-q) \max[W^E - k^m; B]\} \\ + \frac{0.5\delta^2}{1-\delta} \{W + \max[W - k^m; B]\}, \end{split}$$

the present value of staying is given by

$$B + \delta\{(1 - q)W^{E} + q\max[W^{W} - k^{m}; B]\} + \frac{0.5\delta^{2}}{1 - \delta}\{W + \max[W - k^{m}; B]\}.$$

 $^{^{11}}$ Note that the unemployment rate rises in the depressed region among immobile workers since fewer of them receive a job offer in their home region.

¹²Moreover, they find that relative regional unemployment rates in 1975 are highly correlated with relative regional unemployment rates in 1987 in the UK, Japan and Italy and moderately in Germany and Sweden. Interestingly, owner-occupation rates are smaller in the latter two countries.

Moving is optimal if

$$W^{W} - B > k^{m} - \delta(qW^{W} - q * \max[W^{W} - k^{m}, B]) + \delta((1 - q)W^{E} - (1 - q) * \max[W^{E} - k^{m}, B])$$
(17)

Three cases have to be distinguished:

1) For workers satisfying $W^{E} - B > k^{m}$, (17) can be written as

$$W^W - B > k^m - \delta(2q - 1)k^m, \tag{18}$$

2) for workers not satisfying $W^E - B < k^m < W^W - B$, (17) simplifies to

$$W^{W} - B > k^{m} - \delta q k^{m} + \delta (1 - q)(W^{E} - B), \tag{19}$$

3) for workers satisfying $W^E - B < k^m$, (17) can be written as

$$W^{W} - B > \frac{k^{m} + \delta(1 - q)(W^{E} - B)}{1 + \delta q}.$$
 (20)

In all cases, the moving constraint is eased more the higher the wage in the booming region W^W and the larger the shock q is. For workers with a sufficiently low moving barrier (case 1), moving becomes even more attractive than without wage adjustments (compare to equation (10) because of the higher wage paid in West.

4.2 Symmetric Shocks

Proposition 7: An adverse symmetric shock that reduces the job offer arrival rate results in a higher level of unemployment. A positive symmetric shock that increase the length of an employment spell results in an increase in mobility and a reduction in the level of unemployment. More workers move during booms, and the effect is stronger the stronger the boom. *Proof:*

I) Recession

Assume that workers receive a job offer every second period while a job last for one period such that all workers whose job ends become unemployed for at least one period. Note that Proposition 1.1 still holds, i.e. that workers always accept a job offer in the region where they live. Thus, we only have

to consider workers whose first job offer comes from the other city. Moreover, as all workers receive benefits in the period immediately following their employment spell, we only need to compare the expected incomes from rejecting and accepting the job offer in the other region at the time the moving decision has to be made, i.e. in the second period. Since job offers come from either region with equal probability, i.e. q=0.5, the expected income from accepting a job offer in the other region at the time the moving decision is made equals

$$W - k^m + \frac{\delta}{1 - \delta^2} B + \frac{0.5\delta^2}{1 - \delta^2} (W + \max[W - k^m; B])$$

The expected income from rejecting the offer equals

$$B + \frac{\delta}{1 - \delta^2} B + \frac{0.5\delta^2}{1 - \delta^2} (W + \max[W - k^m; B])$$

Therefore a worker accepts the job offer and moves iff

$$W - B > k^m. (21)$$

The length of the recession does not alter the moving decision as long as both regions are affected by the recession in the same way, because an adverse symmetric shock does not alter the moving constraint (1) given that wages remain unchanged. The shock does, however, raise the level of unemployment, because everybody enters the unemployment pool after an employment spell. The expected unemployment duration exactly one period for mobile workers (satisfying (1)) but exceeds 1 period for immobile workers. Consequently, the unemployment rate is higher among immobile workers. If wages fall during recessions, mobility is reduced as the moving constraint (1) is less likely to be satisfied.

II) Booms

Next, consider what happens if jobs last longer than the period between two job offers, i.e. if rate at which job offers arrive increases. In particular, assume that employment lasts for 2 periods. Workers receive a job offer when their job ends or after each period when being unemployed. It is implicitly assumed that workers must fulfill their contract.¹³

¹³Alternatively, we could assume that everybody receives a job offer each period even during employment. But this complicate the analysis as it alters the ex ante probability of not having to move once a job is accepted and would lead to more job changes within a region. After the first period of employment the worker receives a job offer where he

- 1. Permanent shocks
- a) Consider first the effect of a permanent shock on workers who would move in absence of the shock, i.e. those for whom $W k^m > B$. Accepting a job offer from the other region yields an expected income of

$$W - k^m + \frac{\delta}{1 - \delta^2} W + \frac{\delta^2}{1 - \delta^2} (W - 0.5k^m),$$

expected income when rejecting is equal to

$$B + \frac{\delta}{1 - \delta^2} (W - 0.5k^m) + \frac{\delta^2}{1 - \delta^2} W$$

Thus, moving is profitable iff

$$W - B > k^m - \frac{0.5\delta}{1+\delta}k^m \tag{22}$$

The term in square brackets on the right-hand side of equation (22) is positive as $\delta > 0$. Comparing (22) to (1) it is clear, that the expected net benefit from moving rising after a positive permanent shock as mobile workers profit from lower expected moving costs having to move less frequently in order to avoid unemployment.

- b) Workers who do not satisfy (1) but fulfill (22) become mobile and will therefore not become unemployed.
- c) Workers who do not satisfy (22) become unemployment when their job ends. Yet, they are still better off after the shock being employed more often in their home region as jobs lasts twice as long.

If employment lasts t periods, the moving constraint is given by

$$W - B > k^m - \frac{0.5(\delta - \delta^t)}{1 - \delta^t} k^m.$$
 (23)

The coefficient $\frac{(\delta - \delta^t)}{1 - \delta^t}$ is increasing in t.¹⁴ Thus, the stronger the boom, the more mobile workers become.

lives with probability 0.5. It is optimal to accept this offer and quit the current job to be employed for at least two more periods in the same region. The worker rationally rejects an offer from the other region to satisfy the current contract as it keeps the possibility alive that he receives an offer where he lives when the current contract terminates. This happens with probability 0.5. So the ex ante probability of not having to move for the next job is 0.75.

¹⁴9 The first derivative with respect to t is given by $\frac{-\delta^t}{1-\delta^t} \ln \delta$ which is positive since $0 < \delta < 1$.

2. Temporary shocks

If the shock is temporary, say lasts for one period such that only jobs accepted in the first period last for two periods, a worker who accepts a job offer and moves has an expected income equal to

$$W - k^m + \delta W + \frac{0.5\delta^2}{1 - \delta} (W + \max[W - k^m; B]),$$

not moving yields

$$B + 0.5\delta(W + \max[W - k^m; B]) + \frac{0.5\delta^2}{1 - \delta}(W + \max[W - k^m; B]).$$

Moving is profitable if

$$W - B > k^m - 0.5\delta(W - \max[W - k^m; B]). \tag{24}$$

For workers who would move in absence of the shock, the moving constraint becomes

$$W - B > k^m - \delta 0.5k^m. \tag{25}$$

For workers with $W - k^m < B$, the moving constraint becomes

$$W - B > k^m - \delta 0.5(W - B). \tag{26}$$

Immobile workers experience a larger reduction in mobility costs during temporary booms, the larger the difference between their wages and the benefit level (see equation (26)). In addition, the moving barrier is reduced more during a temporary boom than during a permanent boom for all workers (compare (24) and (25) and note that $\delta > \frac{\delta}{1+\delta}$. Opportunity costs of not moving are higher during a temporary boom.

5 Refinements of the Simple Model

So far the analysis has been fairly crude and assumed that high-skill and low-skill workers only differ with respect to wages. This view surely neglects many well established empirical facts of labor markets. There is ample evidence that high-skilled workers receive more wage offers and that employment spells for workers with firm-specific skills and workers in high-wage industries last longer than jobs for low-skilled workers. Two general propositions are derived from the results of the previous section concerning changes in job offer arrival rates and expected length of employment spells. Next, it is analyzed what happens when individual wage offer distributions differ.

5.1 Variation in Job Offer Arrival Rates

<u>Proposition 8:</u> Workers who are in low demand and therefore receive job offers at a lower rate face higher moving barriers. They therefore move less and are unemployed more often. Workers enjoying a higher rate of job offers become more mobile and less likely to become unemployed.

Proof: See the proof of Propositions 6 and 7.

Since high-skilled workers are more likely to receive job offers at a higher rate, high-skilled workers are more mobile, less frequently unemployed and more often observed to move in the data.

5.2 Variations in the Length of Employment Spells

Proposition 9: The longer is the expected length of an employment spell, the more mobile a worker becomes and the less likely he is to become unemployed. Since high-skilled workers, and in particular those with firm-specific skills, have longer expected employment spells with a given firm, they are more mobile and less likely to become unemployed than low-skilled workers.

Proof: See the proof of proposition 7, in particular the argument for positive symmetric shocks.

5.3 Implications when Individual Wage Offer Distributions Differ

It has been assumed so far that individual wages are deterministic (assumption 4). But it is more realistic to assume that an individual worker draws an offer from some distribution. Suppose that the cumulative distribution function of the wage distribution from which a high-skilled worker gets a draw, is given by F(w), while the cumulative distribution function of the wage distribution for a low-skilled worker is defined as G(w). The respective density functions are defined as f(w) and g(w). Assume that the expected wage is H for a high-skilled worker and L for a low-skilled worker. Notice that the model in section 3 is a special case in which F and G are degenerate (since $f(H) = g(L) = \infty$).

Next, define the wage that is required to satisfy the moving constraint as w^o for home-owners and w^r for renters, with $w^o > w^r$. Consider workers receiving their job offer from a firm in the other town. The probabilities that workers accept the offer and move are given by $1 - F(w^r)$ for high-skilled

renters, $1 - F(w^o)$ for high-skilled owners, $1 - G(w^r)$ for low-skilled renters, and $1 - G(w^o)$ for low-skilled owners.

Renters are more likely to move than owners, since $w^o > w^r$ and therefore $1 - F(w^r) > 1 - F(w^o)$ and $1 - G(w^r) > 1 - G(w^o)$ because $F(w^r) < F(w^o)$ and $G(w^r) < G(w^o)$. High-skilled renters are more mobile than low-skilled renters if

$$F(w^r) < G(w^r), (27)$$

while high-skilled owners are more mobile than low-skilled owners if

$$F(w^o) < G(w^o). (28)$$

The condition that makes high-skilled owners more mobile than low-skilled renters is given by

$$1 - F(w^{o}) > 1 - G(w^{r})(F(w^{o}) < G(w^{r}).$$
(29)

Conditions (27) and (28) are always satisfied if the wage distributions for high-skilled and low-skilled differ only in the mean (which is larger for high-skilled) such that F(w) < G(w) for any w in the range of all job offers. Notice that this corresponds to the assumption in previous sections. But other characteristics of the density function can play an important role as well.

<u>Proposition 10:</u> Everything else equal, a rise in the mean of the wage distribution increases the probability of receiving a wage offer that makes moving optimal.

Proof: See the argument above.

<u>Proposition 11:</u> Given that the wage wm that is required to make moving optimal is larger than the expected wage, an increase in the variance of the wage offer distribution tends to increase the probability of receiving an offer exceeding wm, and hence tends to raise mobility.

Proof: If f(w) and $\tilde{f}(w)$ are the same in every respect but the variance, which is bigger for $\tilde{f}(w)$ say, then for values exceeding the mean \bar{w} ,

$$\tilde{F}(w) < F(w),$$

where \tilde{F} and F are the cumulative distribution functions of \tilde{f} and f respectively. Since by assumption $w^m > \bar{w}$,

$$\tilde{F}(w^m) < F(w^m),$$

and job offers that make moving optimal arrive more often.

It is likely that the variance of an individual's wage offer distribution is positively related to his skills, since higher skilled workers have a broader range of jobs to choose from. In addition, if there are inter-industry wage differentials such that some industry x pays wages z percent higher than industry y, then the wage distribution for those earning higher incomes is likely to have a bigger variance. To appreciate this, assume that factor requirements are the same in both industries, then the wage offer distribution for high-skilled have a larger variance than that for less-skilled. 16

A straightforward way to integrate this idea into the simple model is to assume that workers of each skill type can either have a job paying above average wages for the respective skill level – such jobs are called "good" jobs in the analysis that follows – or a job paying below average wages, which are named "bad" jobs. High-skilled workers with a "good" job earn (1+h)H while those having a "bad" job are paid (1-h)H. Similarly, low-skilled can either earn (1+l)L or (1-l)L. If there are as many "good" as "bad" jobs, h^2H^2 and l^2L^2 are the variance of the job offer distribution for high-skilled and low-skilled workers respectively. If l=h, i.e. if remuneration of "good" jobs is 2h percent higher than that of "bad" jobs for both types of workers, the wage offer distribution for workers with a higher mean income also has a larger variance. Assume that workers get a job-offer for either job with equal probability such that expected income is given by H for high-skilled and L for low-skilled. The moving constraint for a high-skilled worker being offered a "good" job is then given by

$$H - B + hH > k^m, (30)$$

for those being offered a "bad" job, it is

$$H - B - hH > k^m, (31)$$

and similarly for low-skilled workers offered a good job the moving constraint is

$$L - B + lL > k^m, (32)$$

¹⁵See Krueger and Summers (1988) for evidence on inter-industry wage differentials.

¹⁶The skewed aggregate wage distribution observed in reality provides support for the hypothesis that the variance of the offer distribution is positively related to income (and hence skills).

¹⁷Jobs in the high-wage industry are hence "good" jobs; equivalently jobs requiring a higher productivity but the same formal qualifications can be interpreted as "good" jobs.

for those being offered a "bad" job, it is

$$L - B - lL > k^m. (33)$$

Suppose for simplicity that l and h are the same. In that case, high-skilled workers who are offered a "good" job are most likely to move. In particular, it is possible that they are the only type of workers who move in absence of any shocks. The condition is

$$(1+h)H - B > k^m > (1-h)H (34)$$

assuming that (1-h)H > (1+l)L.

Note also that in this simple setting the searching constraint derived above is not affected since the one period ahead forecast of income is still given by H for high-skilled and L for low-skilled workers.

6 Interactions between Search and Moving Costs

6.1 A Model with Costly Search

The simple model of section 3 reveals that moving costs reduce workers' willingness to accept jobs – which is reflected in prominent labor market models as a fall in search effectiveness. ¹⁸ This section explains how fixed moving costs interact with search. The model predicts, consistent with empirical findings in the literature, that high-skilled search more and prefer on-the-job search. ¹⁹

So far it has been assumed implicitly that search is costless, since workers automatically receive a job offer at given intervals. Moreover, the simple model does not distinguish between on-the-job search and search from unemployment. It is reasonable to assume that on-the-job search is more expensive than search from unemployment.²⁰

The simple model of section 3 is altered in two ways. First, a worker becomes unemployed for one period after his job contract expires. In each of the following periods, he receives a job offer free of charge. Second, a worker can search for a job while employed to generate one job offer at a fixed cost,

¹⁸See Pissarides (1990), for example.

¹⁹See Pissarides and Wadsworth (1994).

²⁰Unemployed often receive free support from employment agencies; furthermore, employed workers have to take time off to search and, unlike unemployed, incur a loss of income.

 $s.^{21}$ The other assumptions of the simple model still hold.

Proposition 12: High-skilled workers are more likely to search while employed than less-skilled and hence are less likely to become unemployed. The smaller is δ (i.e. the less a worker values the future), the lower is search intensity and the higher is unemployment. The more expensive moving becomes, the less likely are mobile workers to search.

Proof: Note first that the moving constraint in absence of search differs slightly from (1); a non-searching worker becomes unemployed every second period which lowers the net benefit from moving. The discounted stream of income for somebody who is offered a job in the other region and moves is given by

$$W - k^m + \frac{\delta}{1 - \delta^2} B + \frac{0.5\delta^2}{1 - \delta^2} (W + \max[W - k^m; B]),$$

while expected income when rejecting the offer equals

$$B + \frac{0.5\delta}{1 - \delta^2} (W + \max[W - k^m; B]) + \frac{\delta^2}{1 - \delta^2} B,$$
 (35)

and therefore it is optimal to move if

$$W - B > k^m + \frac{0.5\delta}{1+\delta}(W + \max[W - k^m, B]) - \frac{\delta}{1+\delta}B.$$
 (36)

Notice that the term $\frac{0.5\delta}{1+\delta}(W+\max[W-k^m;B])-\frac{\delta}{1+\delta}B$ is always positive, confirming that moving in this setting is generally less attractive than in the simple model

I) Consider now the search decision of workers satisfying (36) who find it always optimal to move when offered a job in the other region. Their expected income when searching only in the first period is given by

$$W - s + \frac{\delta}{1 - \delta^2} W - \frac{0.5\delta}{1 - \delta^2} k^m + \frac{\delta^2}{1 - \delta^2} B,$$

while not searching yields an expected income of

$$W + \frac{\delta}{1 - \delta^2} B + \frac{\delta^2}{1 - \delta^2} W - \frac{0.5\delta^2}{1 - \delta^2} k^m,$$

 $^{^{21}}$ The assumption that search from unemployment is free of charge is not crucial, that is important is that employed search is more expensive. Hence, we scale such that search from unemployment costs nothing; therefore s can be interpreted as the excess cost of employed over unemployed search.

Hence, search is optimal if

$$W - B > \frac{1+\delta}{\delta}s + 0.5k^m. \tag{37}$$

Moreover, it can be shown that a worker satisfying (36) will always search. His expected income when he always searches is given by

$$W - s + \frac{\delta}{1 - \delta}W - \frac{\delta}{1 - \delta}s - \frac{0.5\delta}{1 - \delta}k^m.$$

Always searching is hence preferred to never searching if

$$W - B > +\frac{1+\delta}{\delta}s + 0.5k^m \tag{38}$$

It is apparent that the higher W, the more likely (38) is to be satisfied. A fall in δ , a rise in search costs, s, and higher moving costs all increase the right-hand side of (38) and therefore reduce search activity of workers who are ready to move.

II) For workers who optimally reject a job in the other region the present value of future income when searching only in the first period is given by

$$W - s + \delta(0.5W + 0.5B) + \delta^{2}(0.5B + 0.5(0.5W + 0.5B) + \delta^{3}(.25B + .75(0.5W + 0.5B)) + \dots$$

The present value of future income when not searching is given by:

$$W + \delta B + \delta^2(0.5W + 0.5B) + \delta^3(0.5B + 0.5(0.5W + 0.5B) + \dots$$

Search is hence profitable if

$$W - B > \frac{\delta + 2}{\delta}s. \tag{39}$$

An increase in search costs and a higher discount rate, i.e. lower δ , make searching less attractive. Search costs are more important for immobile workers – the coefficient on search costs in (39) is bigger than the one in (??) – since workers not satisfying the moving constraint only accept half of the job offers they generate and hence their search effort is useless in half of the cases. However, the excess amount of income over benefits, W-B, that is required to make search attractive is lower for immobile workers if

$$k^m > \frac{2s}{\delta}. (40)$$

Note that (40) is always satisfied if the moving constraint is binding but (39) is not, i.e. if

$$\frac{1+\delta}{\delta}s + 0.5k^m > W - B > \frac{\delta+2}{\delta}s,$$

since this implies that

$$k^m > \frac{2+\delta}{\delta - 0.5\delta^2} s,\tag{41}$$

and (40) is satisfied if (41) holds since $\frac{2+\delta}{\delta-0.5\delta^2} > \frac{2}{\delta}$.

Thus, higher moving costs, lower search costs or a lower discount rate δ make mobile workers less likely to search relative to immobile workers. However, the unemployment rate of non-searching mobile workers is lower than the unemployment rate of searching immobile workers as the expected duration of unemployment is longer for immobile workers. For mobile workers who do not search, the expected duration of unemployment is 1 period, while the expected duration for non-searchers is 2 periods. An increase in moving costs has two effects on search intensity. First, it reduces the net gain of search of mobile workers and hence tends to diminish their search activity. This has a negative impact on employment. Second, it increases the proportion of immobile workers. This tends to increase overall search activity but reduces search effectiveness since immobile workers accept only half of the offers. Although the inflow rate into unemployment might fall – it is possible that some mobile workers who did not search become immobile in response to higher moving costs but find it then attractive to search – outflow rates fall even more as unemployment duration increases. Thus, the unemployment rate rises. Both effects have a negative impact on matching and tend to reduce employment. Workers can be a member of one of the following four groups:

- 1. If they satisfy (36) and (37 or 38) they will search, accept any offer and will never become unemployed.
- 2. Workers might satisfy the search condition but do not find it optimal to move. Proportion q of them will become unemployed, where q is the probability of receiving a job offer in the other region.
- 3. Workers are mobile, but do not search. All of them will go through a spell of unem-ployment which lasts exactly for one period, after which they accept a job offer in any region.
- 4. Workers neither move nor search. These workers will become unemployed and the expected duration of unemployment is 2 periods.

6.2 Implications of Shocks

A random temporary shock impacts on search behaviour as workers decide to search before the shock is revealed. A negative (positive) symmetric shock increases (reduces) unemployment among immobile workers. An asymmetric shock increases (decreases) unemployment in the region that is negatively (positively) affected since it reduces (increases) job offers for immobile workers. Mobile workers in the adversely affected region face, on average, higher ex post moving costs, while workers in the booming region gain. The shock has no impact on the employment status of mobile workers. Things are different if workers expect the shock to persist in the future. If q, the probability of getting a job offer in West, is larger than 0.5 only in the next future period, then workers will take into account the higher employment prospects in West. The moving constraint after the first period remains unchanged since it does not matter where one lives 2 periods ahead. However, searching becomes less attractive for workers in the depressed region and more attractive in the booming region. For mobile workers in the depressed (booming) region, the expected benefit from searching falls (rises) by $\delta(q-0.5)k^m$, whereas the expected gain from searching falls (rises) by $\delta(q-0.5)(W-B)$ for immobile workers. The analysis of the previous section revealed that worker in the depressed region become more likely to move if the shock persists for longer. In addition, the value of search increases for workers living in the depressed region because working in the booming region is more attractive and searching increases the chances of finding employment there early. The main insight of the above analysis is that employed search is likely to be optimal for high-skilled workers while it is less likely to be optimal for low-skilled workers. Since it is more profitable for high-skilled workers to engage in costly on-the-job search activity they are less likely to become unemployed. Moreover, moving costs, and therefore the type of housing a worker chooses, influences search behaviour. The result derived in section 3 that moving costs have a negative impact on employment continues to hold. It should be noted that the assumptions about the length of the periods of employment and unemployment are not crucial to derive these implications. In principle, the longer expected employment becomes relative to the period between two job offers, the more likely a worker becomes to search and move. The length of the periods chosen facilitates the illustration of the main implications of the model. The model predicts that higher search costs have a negative impact on employment, a fact that is rather obvious and has received considerable discussion in the literature. Factors that can reduce search costs are, for example, a well functioning personal network, or the access to information. Furthermore, opportunity costs influence employed search behaviour. Workers whose wage largely reflects firm specific skills face higher opportunity costs than workers with more general skills from which they can profit in other firms or industries if they want to leave a firm because they expect their job to end. Since university graduates are likely to have more general skills they face lower opportunity costs of changing jobs and are hence more likely to search and move. The model also implies that immobile workers would concentrate their search activity locally, if this was possible. This limits their alternatives and is likely to lead to inefficient matching which has a negative impact on productive efficiency.

7 Autocorrelation in the Quality of Sequential Job Offers

7.1 Stigmatisation and Insider Forces

In section 3 it has been assumed that workers get job offers for "good" and "bad" jobs with equal probability. This might not be a realistic assumption, however. If unexplained inter-industry wage differentials cause the differences in earnings, it is reasonable to believe that workers already working in the high-wage industry are more likely to remain there because of some insider power like their social network, for example. Likewise, if the wage differentials exist because workers with identical formal qualifications, differ in with respect to other qualities relevant for the job, it is not obvious why the market does not reward them accordingly. One reason could be that workers' qualities are not readily observed; rather firms learn about a worker's true quality over time. Suppose that firms rely on the employment history of a worker as a signal. Workers who had a "good" job before, receive a job offer for a "good" job with probability p, where 0.5 , and a "bad"job with probability 1-p (p depends on the effectiveness of the screening process or the importance of other forces like insider power). The probability is not equal to one, since some workers' qualities might have been misjudged previously.²² Similarly, workers who held a "bad" job previously receive a job offer for a "good" job with probability r and for a "bad" job with probability 1-r, where 0 < r < 0.5. Moreover, assume that unemployment is the worse signal, such that unemployed only receive offers for "bad" jobs. Consider a high-skilled worker with a "good" wage offer in the other region.

 $^{^{22}}$ The probability p refers to the group of workers with a "good" job rather than to individual workers within the group. In fact, the probability of retaining a "good" job for somebody whose qualities were misjudged is lower than p since at least his old employer will not offer him the same job again.

His expected stream of income until the first two periods after accepting is given by

$$(1+h)H - k^m + 0.5\delta\{p(1+h)H + (1-p)(1-h)H\} + 0.5\delta\{p\max[(1+h)H - k^m; B] + (1-p)\max[(1-h)H - k^m; B]\}$$

If the worker does not accept, expected income in the next periods is given by

$$B + 0.5\delta(1-h)H - 0.5\delta p \max[(1-h)H - k^m; B].$$

Note that expected income from the third period onwards is higher if accepting the job offer because upon rejecting, the worker can at best get a "bad" job in the second period and therefore only get a "good" job offer in the third period with probability r. This probability is higher for someone who accepts initially. Hence rejecting is the worse strategy if

$$H - B > k^{m} - hH - \delta phH - \delta 0.5p \max[(1+h)H - k^{m}; B] -0.5\delta p \max[(1-h)H - k^{m}; B] - \delta^{2}v,$$
(42)

where v is the discounted excess value of income from the third period onwards when having accepted initially. Comparing (42) with (30) it is obvious that the opportunity costs of not moving rise for the group of workers with a "good" job. Stigmatisation triggers autocorrelation in the quality level of the job offer sequence. Analogously, the moving condition for a low-skilled worker with a "good" job becomes

$$L - B > k^{m} - lL - \delta p lL - 0.5 \delta p \max[(1 + l)L - k^{m}; B]$$

$$-0.5 \delta p \max[(1 - l)L - k^{m}; B] - \delta^{2} v$$
(43)

For a high-skilled worker being offered a "bad" job in the other region the stream of income in the next two periods is given by

$$(1-h)H - k^m + 0.5\delta\{r(1+h)H + (1-r)(1-h)H\} + 0.5\delta\{r\max[(1+h)H - k^m; B] + (1-r)\max[(1-h)H - k^m; B]\}.$$

If the worker does not accept, expected income in the next periods is given by:

$$B + 0.5\delta(1 - h)H - 0.5\delta p \max[(1 - h)H - k^m; B], \tag{44}$$

The moving constraint becomes

$$H - B > k^{m} + hH - \delta rhH - 0.5\delta r \max[(1 + h)H - k^{m}; B] -0.5\delta r \max[(1 - h)H - k^{m}; B] - \delta^{2}v,$$
(45)

where v is again the excess value of accepting the job offer initially from the third period onwards.

Similarly, for low-skilled workers offered a "bad" job in the other region, the moving constraint becomes

$$L-B > k^{m} + lL - \delta r lL - 0.5 \delta r \max[(1+l)L - k^{m}; B] -0.5 \delta r \max[(1-l)L - k^{m}; B] - \delta^{2} v,$$
(46)

Comparing (45) and (46) to (31) and (33) respectively, it is clear that stigmatisation raises the opportunity cost of not moving for workers with a "bad" job. If stigmatisation is relevant, a given fixed moving cost impedes mobility less. Once again, the effect is strongest for those with highest earnings, which makes this group even more mobile relative to other groups. In addition, a larger variance of the job offer distribution, i.e. a larger h or l, increase the net benefit from moving if job offers depend on the employment history. This potentially explains why young entrants into the labor market are willing to move in order to accept a good job offer. The job accepted today might open up future employment opportunities and thereby raise the mean and variance of the individual job offer distribution. This is especially true for university graduates.

7.2 Search and Stigmatisation

The analysis of this section can also be integrated into the framework analyzed in section 6 which explains why workers with higher earnings search more and prefer employed search. To ease the exposition, it is assumed that workers search only in the first 2 periods. It is clear, however, that searching always is optimal given search is profitable in the first two periods since the third period can be interpreted as the start of a new process.

A high-skilled worker with a "good" job searching in the first two periods has an expected stream of income in the first four periods of

$$(1+h)H - s + \delta[H + 2phH - hH - qk^{m} - s] + \delta^{2}[H + 2p^{2}hH - 2rphH + 2rhH - hH - qk^{m}] + \delta^{3}B$$

The present value of expected future income when not searching is given by

$$(1+h)H + \delta B + \delta^{2}[(1-h)H - qk^{m}] + \delta^{3}B$$
 (47)

Hence, search is profitable if

$$H - B + (2p - 1)hH + \delta[2p^2hH + 2(1 - p)rhH] > \frac{1 + \delta}{\delta}s + qk^m$$
 (48)

Comparing this to the search condition derived in section 2.3, we see that workers with wages above average become even more likely to search since the two extra terms on the left-hand side are positive. Analogously, the search condition for low-skilled workers with a "good" job is derived. It is given by

$$L - B + (2p - 1)lL + \delta[2p^{2}lL + 2(1 - p)rlL] > \frac{1 + \delta}{\delta}s + qk^{m}.$$
 (49)

Searching is profitable for a high-skilled worker with a "bad" job if

$$(1-h)H - B + 2rhH + \delta[2rphH + 2(1-r)rhH] > \frac{1+\delta}{\delta}s + qk^{m}.$$
 (50)

while a low-skilled worker with a "bad" wage offer optimally searches if

$$(1-l)L - B + 2rlL + \delta[2rplL + 2(1-r)rlL] > \frac{1+\delta}{\delta}s + qk^{m}.$$
 (51)

Earning less, (1-h)H instead of H or (1-l)L instead of L, reduces the likelihood of search, as one would predict given the results of section 6. But the effect of stigmatisation has a positive impact on search which is reflected in the two additional terms on the left-hand side of equations (50) which are both positive and hence increase the net benefit of search.

8 Policy Implications

The simple model implies that policies that raise moving costs, reduce mobility, and thereby increase unemployment. The analysis of asymmetric shocks indicates that higher moving costs slow the adjustment process as unemployment among immobile workers in the slump region remains high for a prolonged period. In addition, there are important interactions with search behavior, and increases in mobility costs tend to reinforce other imperfection that can lead to hysteresis effects. The extensions to the simple model generally show that adverse effects of moving costs on unemployment do not only work through the direct channel of fixed moving costs that impede regional mobility, but also through indirect channels as interaction with search behavior or job acceptance behavior (i.e. tend to increase the reservation wage). These effects are largely neglected in the theoretical and empirical literature, but potentially explain the surprising correlation between home ownership rates and the level of unemployment.

A tax deduction, tax benefit or subsidy to moving costs obviously increases mobility and thereby the equilibrium level of unemployment. However, when a policy is tailored to a particular form of housing tenure, substitution effects have to be considered. For example, a subsidy for owning or building a home tends to makes owning more attractive relative to renting. The subsequent rise in the home ownership rate increases moving costs for new owners which might more than offset the initial reduction. Moreover, if such a subsidy is only paid once, for example for the first home bought, future moving costs of owners are not affected by the increase which leads to higher average moving costs in the future as the rate of home owners rises over time due to those who become owners to collect the initial subsidy.

On the other hand, subsidies to the renting sector have to be evaluated carefully. At first sight, rent control would probably judged beneficial for renters as it is designed to keep rents fairly cheap. However, rent control affects the investment decision in the private renting sector. In particular, rent control tends to reduce the return on investment and thus triggers a reduction of supply, either because property is sold off or simply due to attrition. As a result, renters are limited in making their optimal housing consumption choice and given the scarcity in the renting sector, private ownership becomes more attractive. Workers leaving the renting sector are worse off than without controlled rents because they preferred renting when their consumption opportunity were not limited. The loss in utility is reflected in lower mobility. Moving costs rise on average and aggregate unemployment tends to rise given the prediction of the model set out above. However, whether unemployment rises more among owners is ambiguous if the most mobile renters leave the private renting sector first, which is likely since the cost of additional immobility are smallest for them. Moreover, those with higher incomes are more mobile and more likely to be able to afford ownership. In addition, a shrinking private renting sectors has negative effects on mobility of those remaining in the renting sector if rate at which landlords reduce supply exceeds the exit rate from the renting sector. This limits choices of those remaining in the renting sector even further. Consequently, average mobility and therefore employment falls in the renting sector as well.²³

It should always be kept in mind that an increase in the proportion of home owners tends to decrease average mobility in the renting sector as the

²³In Britain, workers at lower incomes have opted to leave the private renting sector for council housing. Mobility of council tenants is lower than mobility of private renters (see Hughes and McCormick (1981)). Council tenants wishing to migrate have to obtain council housing in their destination region and that process does not seem to work well. Hughes and McCormick (1987) argue that council tenants are no less likely than owner-occupiers to wish to migrate, but merely less successful in fulfilling their intentions.

more mobile workers tend to leave the renting sector first. Those who are closer at the margin where home owning is preferred to renting are likely the ones who are affected by changes in policy. Therefore, it would not be surprising to see the level of unemployment increase more in the renting sector when the ownership rate rises.

9 Possible Extensions of the Model

The model developed in this paper can be extended in future work to take account of important features of reality. First, the model neglects the possibility that workers can commute. Commuting is an alternative to moving and hence increases opportunities for workers. Due to congestion costs, commuting costs are a positive function of the number of commuters. Those with lowest moving costs will eventually prefer moving to commuting. The number of commuters is likely to exceed the social optimum because of the externality caused by congestion costs. Moreover, distance is important in reality. To account for this, elements from gravity models might be integrated with the kind of modelling suggested in this paper.

Second, the consumption and investment characteristics of housing and the adjustment of housing prices deserve closer attention. The model establishes that mobility depends on moving costs, but neglects effects of expected capital gains and losses for home-owners. If house prices are expected to rise because the region is hit by a positive asymmetric shock which triggers inmigration and hence an increase in housing demand, owners in the booming region are reluctant to move and sell their house because they would not able to take advantage of the capital gain on their housing investment. Retired or inactive workers in the booming region become less likely to move and fewer housing slots are freed for potential migrants. This tends to slow the adjustment process. Renters (and maybe those owners who can afford to own more than one house) on the other hand can separate their investment and consumption decision. If the consumption and investment decision are separated, there is a freer flow of capital and house price differentials are reduced more quickly.

Changes in housing prices also impact strongly on costs of living, since a large proportion of income is spent on housing, and therefore influence the wedge of consumer and producer prices. This is likely to affect reservation wages and hence wage setting. In addition, consumption demand for other goods is likely to fall if the proportion of income spend on housing rises and hence aggregate demand is dampened. Moreover, fluctuations in housing prices affect different groups in varying degrees and lead to important income

distribution issues.

Third, more attention should be paid to the adjustment process of wages when there is an asymmetric shock as described in section 4.1. It was argued that relative wages fall in the depressed region and shown that this encourages out-migration. However, the determination of wages was not endogenised. Wage differentials are reduced during persistent shocks due to additional in-migration. In addition, lower wages attract more firms to the depressed region and demand is likely to rise in the depressed city and to fall in the booming city. These issues can be analyzed in a model that endogenizes job creation and job destruction. Differences in size, or better in population density, are also likely to impact on the adjustment process. The more densely populated or the bigger the city is, the higher is effective supply for each vacancy. Productivity is likely to be higher due to more efficient job matching. This tends to attract firms to the bigger city.

10 Conclusion

The model explains why high-skilled workers move more and are therefore unemployed less than low-skilled workers. It is consistent with the empirical regularity that higher owner-occupation rates are associated with higher rates of unemployment. Home-owners are less likely to move than renters, ceteris paribus, because their moving costs are higher. But this effect is offset if home-owners earn more since mobility increases with income.

Regional mobility depends on moving costs and on the difference between the value of employment and the value of unemployment. This difference is higher for workers with higher wages. Since wages are a positive function of skills, high-skilled workers are more likely to move regions to find employment, ceteris paribus. The more mobile a worker is, the less likely he is to become unemployed. If mobile workers become unemployed, the expected duration of their unemployment spell is shorter than for immobile workers.

A rise in the value of unemployment relative to the value of employment, e.g. an increase in unemployment benefits or a reduction in wages, reduces mobility and therefore causes unemployment to rise. Similarly, higher moving costs are associated with less mobility. Owners, who incur higher moving costs than renters, are therefore less mobile, ceteris paribus. However, workers with higher wages are more likely to become home-owners as owning is more affordable for them. If high-skilled are more likely to be home-owners, it is possible that the group of home-owners is more mobile than the group of renters.

If the private renting sector shrinks, e.g. because of government interven-

tion or a change in preferences, more mobile workers are likely to leave the renting sector first and hence average mobility of the renting sector and of the owner-occupied sector falls. A higher rate of owner-occupation is therefore associated with more unemployment.

Workers are more likely to move during economy-wide booms because expected moving costs fall. The reduction is larger the higher the wage. Hence, mobility of workers with lowest wages is affected least. Asymmetric shocks tend to increase the number of workers moving to the booming city and reduce the number of workers leaving the booming city. The impact on mobility depends positively on the intensity and persistence of the shock. Mobility is increased most for workers with higher wages. Thus, the model predicts that high-skilled workers are the first to leave a depressed region.

Workers who are willing to move search more effectively. Search effectiveness is an increasing function of mobility. If mobility is reduced, vacancies are filled less efficiently, which is reflected in an outward shift of the Beveridge curve as is observed in European countries.

If job offers follow an autocorrelated process, which could for instance be the result of stigmatisation and asymmetric information concerning individual productivity, workers with higher incomes also face higher opportunity costs of not moving. Moreover, it is suggested that young workers are more mobile if their future income depends on jobs taken presently. In addition, workers whose job offer distribution has a larger variance, ceteris paribus, are more likely to receive a job offer that makes moving profitable.

The choice of housing tenure has important implications for labor market outcomes. Moving costs are hence an important variable that should become part of prominent models of unemployment. Taking account of regional mobility is likely to prove an important step towards explaining the puzzle of European unemployment which, despite a considerable amount of empirical and theoretical work, remains largely unsolved.

A Appendix

Proof that moving to the booming city becomes more attractive the longer the boom is expected to persist:

Suppose it is optimal for the worker in East not to move to West, the booming city, if the asymmetric shock persists only in the next future period. The worker does not satisfy condition 11, i.e. $W - B < k^m - \delta(2q - 1)(W - B)$ for him. If the shock persists one period longer, the worker will not move in

the second period if he had not moved in the first. Consider a worker who receives a job offer in the booming city and expects the shock to persist for the next 2 periods. Not accepting the offer gives an income of

$$B + \delta\{(1-q)W + qB\} + \delta^2\{(1-q)W + qB\} + \delta^3V$$

where V is the expected stream of income discounted to period 3 that is the same in expectation for movers and non-movers. The income when accepting can be calculated by backward induction and is given by

$$W - k^m + \delta \{qW + (1-q)B\} + \delta^2 \{qW + (1-q)B\} + \delta^3 V$$

since $W - B < k^m - \delta(2q - 1)(W - B)$ and W > B. A worker will hence move if

$$W - B > k^m - \delta(2q - 1)(W - B) - \delta^2(2q - 1)(W - B)$$
 (A1)

When the shock persists for just one more future period, the moving constraint was given by

$$W - B > k^m - \delta(2q - 1)(W - B). \tag{14}$$

Comparing (A.1) with (14) it is obvious that moving becomes more likely since the additional term, $\delta^2(2q-1)(W-B)$ in (A.1) is positive.

References

- Blanchard, O. J. and Katz, L. F.: 1997, What We Know and Do Not Know About the Natural Rate of Unemployment, *Journal of Economic Perspectives* **11**(1), 51–72.
- Bover, O., Muellbauer, J. and Murphy, A.: 1989, Housing, Wages and UK Labour Markets, Oxford Bulletin of Economics and Statistics **51**(2), 97–136.
- Contini, B. and Revelli, R.: 1997, Gross flows vs. net flows in the labour market: What is to be learned?, *Labour Economics* 4.
- Davis, S. J., Haltiwanger, J. C. and Schuh, S.: 1996, *Job Creation and Destruction*, MIT Press, Cambridge, MA.

- Evans, P. and McCormick, B.: 1994, The New Pattern of Regional Unemployment: Causes and Policy Significance, *Economic Journal* **104**, 633–647.
- Farber, H. S.: 1999, Mobility and Stability: The Dynamics of Job Change in Labor Markets, in O. Ashenfelter and D. Card (eds), Handbook of Labor Economics, Vol. 3B, North-Holland, Amsterdam.
- Henley, A.: 1998, Residential Mobility, Housing Equity and the Labour Market, *Economic Journal* **108**(447), 414–428.
- Hughes, G. A. and McCormick, B.: 1981, Do Council Housing Policies Reduce Migration between Regions, *Economic Journal* 91, 919–939.
- Hughes, G. A. and McCormick, B.: 1985a, Migration Intentions in the UK, *Economic Journal* **95**, 113–123. Conference Papers Supplement.
- Hughes, G. A. and McCormick, B.: 1985b, An Empirical Analysis of On-the-Job Search and Job Mobility, *The Manchester School* pp. 76–95.
- Hughes, G. A. and McCormick, B.: 1987, "Housing Markets, Unemployment and Labor Market Flexibility in the UK", *European Economic Review* 31, 615–641.
- Layard, R., Nickell, S. J. and Jackman, R.: 1991, *Macroeconomic Performance and the Labour Market*, Oxford University Press, Oxford.
- McCormick, B.: 1997, Regional Unemployment and Labour Mobility in the UK, European Economic Review 41, 581–589.
- Nickell, S. J.: 1997, Unemployment and Labor Market Rigidities: Europe versus North America, *Journal of Economic Perspectives* **11**(3), 55–74.
- Nickell, S. J.: 1998, Unemployment: Questions and some Answers, *Economic Journal* **108**(448), 802–816.
- OECD: 1996, Employment Outlook, OECD, Paris.
- Oswald, A. J.: 1996, A Conjecture on the Explanation for High Unemployment in the Industrialized Nations: Part I. Warwick Economics Research Paper 475.
- Oswald, A. J.: 1997a, Theory of Homes and Jobs. University of Warwick.
- Oswald, A. J.: 1997b, The Missing Piece of the Unemplyment Puzzle. Inaugural Lecture.

- Owen, D. and Green, A.: 1997, Analysis of Geographical Mobility. Working Paper.
- Pissarides, C. A.: 1990, Equilibrium Unemployment, Basil Blackwell, Oxford.
- Pissarides, C. A. and Wadsworth, J.: 1994, On-the-job search: Some Empirical Evidence from Britain, *European Economic Review* **38**(2), 385–401.

IZA Discussion Papers

No	Author(s)	Titel	Area	Date
111	V. Sorm K. Terrell	Sectoral Restructuring and Labor Mobility: A Comparative Look at the Czech Republic	1/4	2/00
112	L. Bellmann T. Schank	Innovations, Wages and Demand for Heterogeneous Labour: New Evidence from a Matched Employer-Employee Data-Set	5	2/00
113	R. Euwals	Do Mandatory Pensions Decrease Household Savings? Evidence for the Netherlands	3	2/00
114	G. Brunello A. Medio	An Explanation of International Differences in Education and Workplace Training	2	2/00
115	A. Cigno F. C. Rosati	Why do Indian Children Work, and is it Bad for Them?	3	2/00
116	C. Belzil	Unemployment Insurance and Subsequent Job Duration: Job Matching vs. Unobserved Heterogeneity	3	2/00
117	S. Bender A. Haas C. Klose	IAB Employment Subsample 1975-1995. Opportunities for Analysis Provided by the Anonymised Subsample	7	2/00
118	M. A. Shields M. E. Ward	Improving Nurse Retention in the British National Health Service: The Impact of Job Satisfaction on Intentions to Quit	5	2/00
119	A. Lindbeck D. J. Snower	The Division of Labor and the Market for Organizations	5	2/00
120	P. T. Pereira P. S. Martins	Does Education Reduce Wage Inequality? Quantile Regressions Evidence from Fifteen European Countries	5	2/00
121	J. C. van Ours	Do Active Labor Market Policies Help Unemployed Workers to Find and Keep Regular Jobs?	4/6	3/00
122	D. Munich J. Svejnar K. Terrell	Returns to Human Capital under the Communist Wage Grid and During the Transition to a Market Economy	4	3/00
123	J. Hunt	Why Do People Still Live in East Germany?	1	3/00
124	R. T. Riphahn	Rational Poverty or Poor Rationality? The Take-up of Social Assistance Benefits	3	3/00
125	F. Büchel J. R. Frick	The Income Portfolio of Immigrants in Germany - Effects of Ethnic Origin and Assimilation. Or: Who Gains from Income Re-Distribution?	1/3	3/00
126	J. Fersterer R. Winter-Ebmer	Smoking, Discount Rates, and Returns to Education	5	3/00

127	M. Karanassou D. J. Snower	Characteristics of Unemployment Dynamics: The Chain Reaction Approach	3	3/00
128	O. Ashenfelter D. Ashmore O. Deschênes	Do Unemployment Insurance Recipients Actively Seek Work? Evidence From Randomized Trials in Four U.S. States	6	3/00
129	B. R. Chiswick M. E. Hurst	The Employment, Unemployment and Unemployment Compensation Benefits of Immigrants	1/3	3/00
130	G. Brunello S. Comi C. Lucifora	The Returns to Education in Italy: A New Look at the Evidence	5	3/00
131	B. R. Chiswick	Are Immigrants Favorably Self-Selected? An Economic Analysis	1	3/00
132	R. A. Hart	Hours and Wages in the Depression: British Engineering, 1926-1938	7	3/00
133	D. N. F. Bell R. A. Hart O. Hübler W. Schwerdt	Paid and Unpaid Overtime Working in Germany and the UK	1	3/00
134	A. D. Kugler G. Saint-Paul	Hiring and Firing Costs, Adverse Selection and Long-term Unemployment	3	3/00
135	A. Barrett P. J. O'Connell	Is There a Wage Premium for Returning Irish Migrants?	1	3/00
136	M. Bräuninger M. Pannenberg	Unemployment and Productivity Growth: An Empirical Analysis within the Augmented Solow Model	3	3/00
137	JSt. Pischke	Continuous Training in Germany	5	3/00
138	J. Zweimüller R. Winter-Ebmer	Firm-specific Training: Consequences for Job Mobility	1	3/00
139	R. A. Hart Y. Ma	Wages, Hours and Human Capital over the Life Cycle	1	3/00
140	G. Brunello S. Comi	Education and Earnings Growth: Evidence from 11 European Countries	2/5	4/00
141	R. Hujer M. Wellner	The Effects of Public Sector Sponsored Training on Individual Employment Performance in East Germany	6	4/00
142	J. J. Dolado F. Felgueroso J. F. Jimeno	Explaining Youth Labor Market Problems in Spain: Crowding-Out, Institutions, or Technology Shifts?	3	4/00
143	P. J. Luke M. E. Schaffer	Wage Determination in Russia: An Econometric Investigation	4	4/00
144	G. Saint-Paul	Flexibility vs. Rigidity: Does Spain have the worst of both Worlds?	1	4/00

145	MS. Yun	Decomposition Analysis for a Binary Choice Model	7	4/00
146	T. K. Bauer J. P. Haisken-DeNew	Employer Learning and the Returns to Schooling	5	4/00
147	M. Belot J. C. van Ours	Does the Recent Success of Some OECD Countries in Lowering their Unemployment Rates Lie in the Clever Design of their Labour Market Reforms?	3	4/00
148	L. Goerke	Employment Effects of Labour Taxation in an Efficiency Wage Model with Alternative Budget Constraints and Time Horizons	3	5/00
149	R. Lalive J. C. van Ours J. Zweimüller	The Impact of Active Labor Market Programs and Benefit Entitlement Rules on the Duration of Unemployment	3/6	5/00
150	J. DiNardo K. F. Hallock JSt. Pischke	Unions and the Labor Market for Managers	7	5/00
151	M. Ward	Gender, Salary and Promotion in the Academic Profession	5	5/00
152	J. J. Dolado F. Felgueroso J. F. Jimeno	The Role of the Minimum Wage in the Welfare State: An Appraisal	3	5/00
153	A. S. Kalwij M. Gregory	Overtime Hours in Great Britain over the Period 1975-1999: A Panel Data Analysis	3	5/00
154	M. Gerfin M. Lechner	Microeconometric Evaluation of the Active Labour Market Policy in Switzerland	6	5/00
155	J. Hansen	The Duration of Immigrants' Unemployment Spells: Evidence from Sweden	1/3	5/00
156	C. Dustmann F. Fabbri	Language Proficiency and Labour Market Performance of Immigrants in the UK	1	5/00
157	P. Apps R. Rees	Household Production, Full Consumption and the Costs of Children	7	5/00
158	A. Björklund T. Eriksson M. Jäntti O. Raaum E. Österbacka	Brother Correlations in Earnings in Denmark, Finland, Norway and Sweden Compared to the United States	5	5/00
159	P J. Jost M. Kräkel	Preemptive Behavior in Sequential Tournaments	5	5/00
160	M. Lofstrom	A Comparison of the Human Capital and Signaling Models: The Case of the Self-Employed and the Increase in the Schooling Premium in the 1980's	5	6/00
161	V. Gimpelson D. Treisman G. Monusova	Public Employment and Redistributive Politics: Evidence from Russia's Regions	4	6/00

162	C. Dustmann M. E. Rochina- Barrachina	Selection Correction in Panel Data Models: An Application to Labour Supply and Wages	6	6/00
163	R. A. Hart Y. Ma	Why do Firms Pay an Overtime Premium?	5	6/00
164	M. A. Shields S. Wheatley Price	Racial Harassment, Job Satisfaction and Intentions to Quit: Evidence from the British Nursing Profession	5	6/00
165	P. J. Pedersen	Immigration in a High Unemployment Economy: The Recent Danish Experience	1	6/00
166	Z. MacDonald M. A. Shields	The Impact of Alcohol Consumption on Occupational Attainment in England	5	6/00
167	A. Barrett J. FitzGerald B. Nolan	Earnings Inequality, Returns to Education and Immigration into Ireland	5	6/00
168	G. S. Epstein A. L. Hillman	Social Harmony at the Boundaries of the Welfare State: Immigrants and Social Transfers	3	6/00
169	R. Winkelmann	Immigration Policies and their Impact: The Case of New Zealand and Australia	1	7/00
170	T. K. Bauer K. F. Zimmermann	Immigration Policy in Integrated National Economies	1	7/00
171	C. Dustmann F. Windmeijer	Wages and the Demand for Health – A Life Cycle Analysis	5	7/00
172	D. Card	Reforming the Financial Incentives of the Welfare System	3	7/00
173	D. S. Hamermesh	Timing, Togetherness and Time Windfalls	5	7/00
174	E. Fehr JR. Tyran	Does Money Illusion Matter? An Experimental Approach	7	7/00
175	M. Lofstrom	Self-Employment and Earnings among High- Skilled Immigrants in the United States	1	7/00
176	O. Hübler W. Meyer	Industrial Relations and the Wage Differentials between Skilled and Unskilled Blue-Collar Workers within Establishments: An Empirical Analysis with Data of Manufacturing Firms	5	7/00
177	B. R. Chiswick G. Repetto	Immigrant Adjustment in Israel: Literacy and Fluency in Hebrew and Earnings	1	7/00
178	R. Euwals M. Ward	The Renumeration of British Academics	5	7/00
179	E. Wasmer P. Weil	The Macroeconomics of Labor and Credit Market Imperfections	2	8/00
180	T. K. Bauer I. N. Gang	Sibling Rivalry in Educational Attainment: The German Case	5	8/00

181	E. Wasmer Y. Zenou	Space, Search and Efficiency	2	8/00
182	M. Fertig C. M. Schmidt	Discretionary Measures of Active Labor Market Policy: The German Employment Promotion Reform in Perspective	6	8/00
183	M. Fertig C. M. Schmidt	Aggregate-Level Migration Studies as a Tool for Forecasting Future Migration Streams	1	8/00
184	M. Corak B. Gustafsson T. Österberg	Intergenerational Influences on the Receipt of Unemployment Insurance in Canada and Sweden	3	8/00
185	H. Bonin K. F. Zimmermann	The Post-Unification German Labor Market	4	8/00
186	C. Dustmann	Temporary Migration and Economic Assimilation	1	8/00
187	T. K. Bauer M. Lofstrom K. F. Zimmermann	Immigration Policy, Assimilation of Immigrants and Natives' Sentiments towards Immigrants: Evidence from 12 OECD-Countries	1	8/00
188	A. Kapteyn A. S. Kalwij A. Zaidi	The Myth of Worksharing	5	8/00
189	W. Arulampalam	Is Unemployment Really Scarring? Effects of Unemployment Experiences on Wages	3	8/00
190	C. Dustmann I. Preston	Racial and Economic Factors in Attitudes to Immigration	1	8/00
191	G. C. Giannelli C. Monfardini	Joint Decisions on Household Membership and Human Capital Accumulation of Youths: The role of expected earnings and local markets	5	8/00
192	G. Brunello	Absolute Risk Aversion and the Returns to Education	5	8/00
193	A. Kunze	The Determination of Wages and the Gender Wage Gap: A Survey	5	8/00
194	A. Newell F. Pastore	Regional Unemployment and Industrial Restructuring in Poland	4	8/00
195	F. Büchel A. Mertens	Overeducation, Undereducation, and the Theory of Career Mobility	5	9/00
196	J. S. Earle K. Z. Sabirianova	Equilibrium Wage Arrears: A Theoretical and Empirical Analysis of Institutional Lock-In	4	9/00
197	G. A. Pfann	Options to Quit	1	9/00

198	M. Kreyenfeld C. K. Spiess G. G. Wagner	A Forgotten Issue: Distributional Effects of Day Care Subsidies in Germany	3	9/00
199	H. Entorf	Rational Migration Policy Should Tolerate Non- Zero Illegal Migration Flows: Lessons from Modelling the Market for Illegal Migration	1	9/00
200	T. Bauer G. S. Epstein I. N. Gang	What are Migration Networks?	1	9/00
201	T. J. Dohmen G. A. Pfann	Worker Separations in a Nonstationary Corporate Environment	1	9/00
202	P. Francois J. C. van Ours	Gender Wage Differentials in a Competitive Labor Market: The Household Interaction Effect	5	9/00
203	J. M. Abowd F. Kramarz D. N. Margolis T. Philippon	The Tail of Two Countries: Minimum Wages and Employment in France and the United States	5	9/00
204	G. S. Epstein	Labor Market Interactions Between Legal and Illegal Immigrants	1	10/00
205	A. L. Booth M. Francesconi J. Frank	Temporary Jobs: Stepping Stones or Dead Ends?	1	10/00
206	C. M. Schmidt R. Baltussen R. Sauerborn	The Evaluation of Community-Based Interventions: Group-Randomization, Limits and Alternatives	6	10/00
207	C. M. Schmidt	Arbeitsmarktpolitische Maßnahmen und ihre Evaluierung: eine Bestandsaufnahme	6	10/00
208	J. Hartog R. Winkelmann	Dutch Migrants in New Zealand: Did they Fare Well?	1	10/00
209	M. Barbie M. Hagedorn A. Kaul	Dynamic Efficiency and Pareto Optimality in a Stochastic OLG Model with Production and Social Security	3	10/00
210	T. J. Dohmen	Housing, Mobility and Unemployment	1	11/00